Author's response to reviews

Title: Distraction osteogenesis in a severe mandibular deficiency

Authors:

Kerim Ortakoglu (kortakoglu@yahoo.com)
Seniz Karacay (senkaracay@yahoo.com)
Metin Sencimen (metinsencimen@yahoo.com)
Erol Akin (erol1akin@hotmail.com)
Aykut H Ozyigit (aykutozyigit@yahoo.com)
Osman Bengi (osmanbengi@hotmail.com)

Version: 2 Date: 28 October 2006

Author's response to reviews:

The title is changed according to the recommendadions
NV is made clear by nasion vertical.
Total treatment time is explained as 14 months.
The terms such as presurgical and postsurgical is changed to pretreatment and posttreatment.
Referance planes for superimpositions are stated as PM and NOA
Relaps is corrected as relapse
Table headings are corrected
Number of figures are reduced.